\begin{table}%t2
\caption{\label{table-identifier}Astrometry, photometry and membership assignments are presented for those M~34 stars for which photometric periods are derived in the Lowell campaign.}
\par
%\centering
\par
{
\begin{tabular}{rcclcrrrrrll}
\hline\hline \noalign{\smallskip}
Internal\tablefootmark{a} & RA\tablefootmark{b} & Dec\tablefootmark{b} & JP 96\tablefootmark{c} & 
$V$\tablefootmark{d} &  Verr\tablefootmark{d} &  $B-V$\tablefootmark{d}  & $B$-Verr\tablefootmark{d} & $V-I$\tablefootmark{d}  & $V$-Ierr\tablefootmark{d}  & Phot\tablefootmark{e} & RV\tablefootmark{e} \\ 
Identifier    & (J2000) & (J2000) & & & & & & & & Mem & Mem \\
\hline
F3\_0172                  & 02 41 05.141 & +42 56 43.202 & JP 49  & 14.584 & 0.017 & 0.881 & 0.023 & 0.956 & 0.024 & Yes &  Yes-sing \\
F3\_0176                  & 02 41 36.611 & +42 50 20.288 & JP 167 & 14.659 & 0.009 & 0.888 & 0.013 & 1.051 & 0.013 & Yes &  Yes-sing \\
F3\_0215                  & 02 41 00.981 & +42 52 46.695 & JP 41  & 14.918 & 0.010 & 0.944 & 0.015 & 1.039 & 0.014 & Yes &  Yes-sing \\
F3\_0258                  & 02 42 02.552 & +42 51 51.721 & JP 289 & 15.236 & 0.009 & 1.017 & 0.013 & 1.118 & 0.013 & Yes &  Yes-sing \\
F3\_0306                  & 02 41 57.472 & +43 00 26.530 & JP 265 & 15.503 & 0.014 & 1.061 & 0.019 & 1.262 & 0.027 & Yes &  Yes-bin  \\
F3\_0320                  & 02 41 37.557 & +42 57 21.593 & JP 172 & 15.664 & 0.017 & 1.116 & 0.025 & 1.237 & 0.024 & Yes &           \\
F3\_0383                  & 02 40 49.882 & +42 55 31.520 &        & 16.179 & 0.017 & 1.230 & 0.025 & 1.424 & 0.024 & Yes &  Yes-sing \\
F3\_0430                  & 02 41 50.490 & +42 58 10.016 &        & 16.424 & 0.013 & 1.309 & 0.019 & 1.505 & 0.018 & Yes &  Yes-sing \\
F3\_0469                  & 02 40 49.090 & +42 48 20.648 &        & 16.604 & 0.018 & 1.353 & 0.030 & 1.550 & 0.025 & Yes &           \\
F3\_0485                  & 02 41 06.638 & +42 48 22.011 &        & 16.732 & 0.010 & 1.355 & 0.016 & 1.619 & 0.014 & Yes &           \\
F3\_0487                  & 02 40 44.889 & +42 53 47.221 &        & 16.818 & 0.020 & 1.348 & 0.030 & 1.701 & 0.027 & Yes &           \\
F3\_0600                  & 02 41 18.447 & +42 58 21.342 &        & 17.394 & 0.020 & 1.522 & 0.035 & 1.863 & 0.027 & Yes &           \\
F3\_0664                  & 02 41 20.454 & +42 58 52.278 &        & 17.862 & 0.022 & 1.511 & 0.057 & 1.999 & 0.029 & Yes &           \\
F4\_0136                  & 02 41 47.384 & +42 43 38.525 & JP 213 & 13.151 & 0.010 & 0.614 & 0.014 & 0.718 & 0.015 & Yes &  Yes-sing \\
F4\_0169                  & 02 41 04.952 & +42 46 51.343 & JP 50  & 13.510 & 0.010 & 0.668 & 0.014 & 0.786 & 0.014 & Yes &  Yes-sing \\
F4\_0194                  & 02 41 28.142 & +42 38 37.095 & JP 133 & 13.662 & 0.010 & 0.722 & 0.014 & 0.805 & 0.015 & Yes &  Yes-sing \\
F4\_0234                  & 02 41 33.429 & +42 42 11.625 & JP 148 & 13.959 & 0.010 & 0.768 & 0.014 & 0.844 & 0.014 & Yes &  Yes-sing \\
F4\_0303                  & 02 41 49.932 & +42 37 13.906 & JP 227 & 14.366 & 0.009 & 0.837 & 0.013 & 0.912 & 0.013 & Yes &  Yes-sing \\
F4\_0317                  & 02 41 49.354 & +42 36 37.613 & JP 224 & 14.410 & 0.010 & 0.875 & 0.014 & 0.959 & 0.014 & Yes &  Yes-sing \\
F4\_0327                  & 02 41 44.167 & +42 46 07.333 & JP 199 & 14.483 & 0.009 & 0.883 & 0.013 & 0.941 & 0.013 & Yes &  Yes-sing \\
F4\_0335                  & 02 41 35.247 & +42 41 02.323 & JP 158 & 14.546 & 0.010 & 0.860 & 0.014 & 0.994 & 0.014 & Yes &  Yes-bin  \\
F4\_0450                  & 02 41 23.098 & +42 40 15.954 & JP 113 & 14.960 & 0.010 & 0.965 & 0.014 & 1.058 & 0.014 & Yes &  Yes-sing \\
F4\_0667                  & 02 41 06.187 & +42 46 20.783 & JP 52  & 15.623 & 0.009 & 1.100 & 0.013 & 1.233 & 0.013 & Yes &  Yes-bin  \\
F4\_0695                  & 02 41 42.922 & +42 33 13.910 & JP 197 & 15.643 & 0.021 & 1.156 & 0.030 & 1.289 & 0.029 & Yes &  Yes-sing \\
F4\_0730                  & 02 40 49.647 & +42 46 54.799 & JP 18  & 15.741 & 0.017 & 1.121 & 0.025 & 1.257 & 0.024 & Yes &           \\
F4\_0736                  & 02 40 48.542 & +42 39 25.716 &        & 15.844 & 0.021 & 1.078 & 0.028 & 1.328 & 0.029 & Yes &  Yes-bin  \\
F4\_0803                  & 02 41 50.301 & +42 44 37.371 & JP 229 & 16.025 & 0.010 & 1.234 & 0.014 & 1.373 & 0.013 & Yes &  Yes-sing \\
F4\_0804                  & 02 41 48.974 & +42 39 59.680 &        & 16.034 & 0.010 & 1.176 & 0.015 & 1.378 & 0.014 & Yes &           \\
F4\_0852                  & 02 41 53.246 & +42 35 26.394 &        & 16.294 & 0.009 & 1.225 & 0.013 & 1.634 & 0.013 & Yes &  Yes-bin  \\
F4\_0942                  & 02 41 46.371 & +42 32 32.128 &        & 16.341 & 0.020 & 1.268 & 0.029 & 1.498 & 0.028 & Yes &           \\
F4\_1007                  & 02 40 50.752 & +42 41 19.844 &        & 16.839 & 0.022 & 1.510 & 0.033 & 1.901 & 0.034 & Yes &           \\
F4\_1020                  & 02 40 36.693 & +42 41 32.500 &        & 16.608 & 0.021 & 1.330 & 0.031 & 1.584 & 0.030 & Yes &           \\
F4\_1055                  & 02 40 42.791 & +42 38 59.153 &        & 16.639 & 0.021 & 1.342 & 0.031 & 1.600 & 0.029 & Yes &           \\
F4\_1147                  & 02 41 43.854 & +42 45 07.859 &        & 17.009 & 0.010 & 1.406 & 0.017 & 1.749 & 0.013 & Yes &           \\
F4\_1315                  & 02 40 53.997 & +42 38 48.513 &        & 17.347 & 0.022 & 1.441 & 0.036 & 1.815 & 0.030 & Yes &           \\
F4\_1357                  & 02 41 44.198 & +42 35 36.074 &        & 17.476 & 0.011 & 1.450 & 0.021 & 1.911 & 0.015 & Yes &           \\
F4\_1369                  & 02 41 51.600 & +42 29 50.864 &        & 17.496 & 0.015 & 1.513 & 0.027 & 1.934 & 0.021 & Yes &           \\
F4\_1392                  & 02 41 38.851 & +42 43 01.148 &        & 17.654 & 0.011 & 1.504 & 0.025 & 1.963 & 0.015 & Yes &           \\
F4\_1566                  & 02 41 31.131 & +42 41 14.283 &        & 18.132 & 0.012 & 1.549 & 0.032 & 2.137 & 0.016 & Yes &           \\
F4\_1617                  & 02 41 26.292 & +42 30 15.124 &        & 18.165 & 0.027 & 1.570 & 0.064 & 2.178 & 0.035 & Yes &           \\
F4\_1793                  & 02 41 20.042 & +42 39 23.652 &        & 18.606 & 0.015 & 1.393 & 0.063 & 2.285 & 0.019 & Yes &           \\
F4\_1925$^{f}$            & 02 41 21.947 & +42 32 32.637 &        & 19.040 & 0.037 & 1.577 & 0.045 & 2.350 & 0.045 & Yes &          \\
F4\_2226$^{f}$            & 02 40 31.108 & +42 40 25.105 &        & 19.659 & 0.067 & 1.625 & 0.071 & 2.626 & 0.071 & Yes &           \\ 
\hline
F3\_0071                  & 02 40 44.422 & +42 54 04.516 &        & 12.855 & 0.019 & 1.047 & 0.025 & 1.153 & 0.026 &  No &           \\
F3\_0121                  & 02 41 47.414 & +42 50 09.223 & JP 212 & 13.741 & 0.009 & 1.248 & 0.013 & 1.354 & 0.013 &  No &           \\
F3\_0163                  & 02 41 55.003 & +42 52 53.485 & JP 258 & 14.440 & 0.008 & 0.900 & 0.011 & 0.986 & 0.012 & Yes &  No       \\
F3\_0165                  & 02 41 51.794 & +42 46 20.473 & JP 244 & 14.419 & 0.008 & 0.865 & 0.011 & 0.956 & 0.011 & Yes &  No       \\
F3\_0388                  & 02 41 38.227 & +42 55 43.112 & JP 174 & 15.952 & 0.010 & 0.975 & 0.015 & 1.076 & 0.022 &  No &           \\
F3\_0413                  & 02 41 48.478 & +42 49 33.840 & JP 218 & 16.242 & 0.010 & 1.143 & 0.015 & 1.412 & 0.014 & Yes &  No       \\
F3\_0437                  & 02 40 48.970 & +42 56 12.434 &        & 16.022 & 0.018 & 0.647 & 0.025 & 0.748 & 0.025 &  No &           \\
F3\_0692                  & 02 40 57.994 & +42 49 44.614 &        & 17.136 & 0.010 & 0.857 & 0.016 & 1.061 & 0.015 &  No &           \\
F3\_2310                  & 02 40 44.085 & +42 53 58.059 &        & 99.999 & 9.999 & 9.999 & 9.999 & 9.999 & 9.999 &     &           \\
F4\_0167                  & 02 41 48.825 & +42 31 31.702 & JP 223 & 13.502 & 0.021 & 0.938 & 0.026 & 1.030 & 0.031 &  No &           \\
F4\_0404                  & 02 41 27.663 & +42 35 42.306 & JP 131 & 14.730 & 0.010 & 0.852 & 0.014 & 0.932 & 0.014 & Yes &  No       \\
F4\_0515                  & 02 41 27.188 & +42 43 41.881 & JP 129 & 15.111 & 0.010 & 0.903 & 0.014 & 0.977 & 0.014 &  No &           \\
\hline
\end{tabular}}
\tablefoot {\tablefoottext{a}{Identifier format is XX\_YYYY, where ``XX'' are representations F3 and F4 for targets detected on M~34 western Field~3 or western Field~4 images, and ``YYYY'' are internal catalog numbers.}\\
\tablefoottext{b}{Astrometric data are derived from coordinate matches in the SuperCosmos survey catalogs (Hambly et~al. \cite{Hamb01a}$-$c).}\\
\tablefoottext{c}{JP 96 identifiers are for photometric variables which are correlated with Jones \& Prosser (\cite{Jone96}) sources.} \\
\tablefoottext{d}{{\it BVI} photometric data are taken from the Indiana Group survey (\cf Sect.~\ref{indiana}).} \\
\tablefoottext{e}{Membership assignments are based upon whether a target star lies on (or close~to) the $V/B-V$ and/or $V/V-I$ main sequence loci (see Fig.~\ref{CMDs}) or whether a target star has a radial velocity consistent with the cluster's systemic velocity (Meibom et~al., in~prep.; see also Sect.~\ref{extantOBS} and Meibom et~al. \cite{Meib06}).} \\
\tablefoottext{f}{$B-V$ colours are determined using a field star $V-I$ to $B-V$~relationship (Caldwell et~al. \cite{Cald93}), assuming $E_{V-I}$~$=$ $1.25$~$\times$ $E_{B-V}$, and $E_{B-V}$~= 0.07 (Canterna et~al. \cite{Cant79}).}}
\end{table}